Author’s response to reviews

Title: Validation of a 5-item tool to measure patient assessment of clinician compassion in the emergency department

Authors:

Praveen Sabapathi (Sabapathi-Praveen1@cooperhealth.edu)

Michael Roberts (michaelrob@pcom.edu)

Brian Fuller (fullerb@wustl.edu)

Michael Puskarich (Michael.Puskarich@hcmed.org)

Christopher Jones (jones-christopher@cooperhealth.edu)

Hope Kilgannon (kilgannon-hope@cooperhealth.edu)

Valerie Braz (braz-valerie@cooperhealth.edu)

Christina Creel-Bulos (creelbulosm@wustl.edu)

Nathaniel Scott (Nathaniel.Scott@hcmed.org)

Kristina Tester (teste043@umn.edu)

Anthony Mazzarelli (mazzarelli-anthony@cooperhealth.edu)

Stephen Trzeciak (trzeciak-stephen@cooperhealth.edu)

Brian Roberts (roberts-brian-w@cooperhealth.edu)

Version: 1 Date: 15 Aug 2019

Author’s response to reviews:

We apologize for the accidental self-plagiarism, which was identified by the editorial staff. We have modified our text to include less overlap, as well as have provided additional citations in sections where the methodology could not be reworded. Thank you for the opportunity to re-submit and we are happy to make additional changes to the text should the editors feel it is warranted.